Author's response to reviews

Title: Characteristics of neonates with culture-proven bloodstream infection who have low levels of C-reactive protein (<10 mg/L)

Authors:

Mei-Yin Lai (lmi818@msn.com)
Ming-Horng Tsai (mingmin.tw@yahoo.com.tw)
Ming-Chou Chiang (cmc123@cgmh.org.tw)
Reyin Lien (reyinl@cgmh.org.tw)
Ren-Huei Fu (rkenny@cgmh.org.tw)
Hsuan-Rong Huang (gbonbon@cgmh.org.tw)
Shih-Ming Chu (kz6479@cgmh.org.tw)
Jen-Fu Hsu (jeff0724@gmail.com)
Chiang-Wen Lee (cwlee@gw.cgust.edu.tw)

Version: 5 Date: 17 July 2015

Author's response to reviews: see over
RE: MS# 5077672091640319
Title: Characteristics of neonates with culture-proven bloodstream infection who have low levels of C-reactive protein (≤ 10 mg/L)

Dear Editor,

Thank you for your appreciated comments on our manuscript. We had the manuscript revised, mostly according to the reviewers’ suggestions. We will respond in detail to the reviewers’ comments. The replies for the reviewers’ criticisms are as followings. We hope this revised version can be acceptable.

Best regards,

Jen-Fu Hsu, MD and Shih-Ming Chu, MD
Chief, Division of Pediatric Neonatology, Department of Pediatrics
Chang Gung Memorial Hospital at Linkou
Kweishan, Taoyuan, Taiwan

Editor’s comment:

"I think the paper can be accepted. It could benefit from some copy-editing to clean-up a few small grammatical errors and to increase the ease of reading."

Editorial request:

1. Copyedit:
We recommend that you copyedit the paper to improve the style of written English.

Reply:
Thank you for your instructive advice. We have the language in our manuscript edited by a native-English speaker with scientific expertise. This service is from the international English Editing Company (The proof is available if requested), thank you.